Author’s response to reviews

Title: Lynch Syndrome from a surgeon perspective: retrospective study of clinical impact of mismatch repair protein expression analysis in colorectal cancer patients less than 50 years old

Authors:

Gian Luca Baiocchi (baiocchi@med.unibs.it)
Nazario Portolani (portolan@med.unibs.it)
William Vermi (vermi@med.unibs.it)
Carla Baronchelli (carla.baronchelli@spedalicivili.brescia.it)
Federico Gheza (federicogheza@virgilio.it)
Claudio Zogno (c.zogno@libero.it)
Alessandro Scaglia (scaglia.ale@hotmail.com)
Eleonora Marchina (marchina@med.unibs.it)
Guido Tiberio (guido.tiberio@libero.it)
Stefano Maria Giulini (giulin@med.unibs.it)

Version: 4 Date: 28 December 2013

Author’s response to reviews: see over
Many thanks for Your suggestion. Indeed, a formal and written consent for manuscript entirely.

Thus, we have actually completely removed the table from the final version of the publication was not obtained for all the patients, some of which were really old.

Could we therefore ask whether consent to publish this information was identifiable. Could we therefore ask whether consent to publish this information was obtained from these patients? If so please include a statement to this effect in your manuscript. Please note that we retain the right to request documentary proof of this consent to publish at any point. If not, then we would ask you to please group this data so as to render the subjects non-identifiable, or to remove this table from your manuscript entirely.

Many thanks for Your suggestion. Indeed, a formal and written consent for publication was not obtained for all the patients, some of which were really old. Thus, we have actually completely removed the table from the final version of the manuscript.

While we appreciate that improvements have been made in the quality of English in your manuscript, we still feel that it would benefit from some further checking and corrections. For example, the word 'history' is misspelt in table 2.

A further revision by Federico Gheza, one of the participating Authors, which spent recently 1 year of his career in Chicago, USA, has actually been done.
• Please remove your copyediting letter from your additional files.
  
  OK this has actually been done

• Please ensure that your Authors’ Contributions section is formatted in accordance with our formatting guidelines. Specifically, authors should be referred to by their initials in this section. Further detail on our formatting guidelines can be found through the following link:
  http://www.biomedcentral.com/bmcsurg/authors/instructions/researcharticle#preparing-main-manuscript

  OK this has actually been done

  Many thanks for Your attention

Dr. Gian Luca Baiocchi
On behalf of all the Authors